



University of Groningen

# LOCATION AND ENTREPRENEURSHIP

Modrego, Felix; McCann, Philip; Foster, William E.; Olfert, M. Rose

Published in: Journal of Regional Science

DOI: 10.1111/jors.12323

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2017

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Modrego, F., McCann, P., Foster, W. E., & Olfert, M. R. (2017). LOCATION AND ENTREPRENEURSHIP: INSIGHTS FROM A SPATIALLY-EXPLICIT OCCUPATIONAL CHOICE MODEL WITH AN APPLICATION TO CHILE. Journal of Regional Science, 57(4), 669-697. https://doi.org/10.1111/jors.12323

Copyright Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

#### Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.

JOURNAL OF REGIONAL SCIENCE, VOL. 57, NO. 4, 2017, pp. 669-697

# LOCATION AND ENTREPRENEURSHIP: INSIGHTS FROM A SPATIALLY-EXPLICIT OCCUPATIONAL CHOICE MODEL WITH AN APPLICATION TO CHILE

### Félix Modrego

IDEAR, Department of Economics, Universidad Católica del Norte, Avenida Angamos 0610, Antofagasta, Chile. E-mail: fmodrego@ucn.cl

# Philip McCann

Faculty of Spatial Sciences, Department of Economic Geography, University of Groningen, PO Box 800, 9700AV, Groningen, the Netherlands. E-mail: P.Mccann@rug.nl

# William E. Foster

Department of Agricultural Economics, Pontificia Universidad Católica de Chile; and Millennium Nucleus Center for Socioeconomic Impact of Environmental Policies (CESIEP), Vicuña Mackenna 4860, Macul, Santiago, Chile. E-mail: wfoster@uc.cl

# **M. Rose Olfert**

Johnson-Shoyama Graduate School of Public Policy, University of Saskatchewan, 101 Diefenbaker Place, Saskatoon, SK S7N 5B8, Canada. E-mail: rose.olfert@usask.ca

**ABSTRACT.** Occupational choice and heterogeneous managerial ability enter a spatial Dixit-Stiglitz setting, linking location, wages and regional entrepreneurship rates. Market potential has a positive partial effect and wages a negative partial effect on the regional supply of entrepreneurs, both balancing in equilibrium with endogenous wages. Market potential increases profits, but also the opportunity cost of entrepreneurship. In the long-run equilibrium with perfect mobility, the cut-off level of ability determining selection into entrepreneurship will be the same across regions; moreover, regional differences in entrepreneurship rates depend only in differences in average fixed costs of firms. An empirical application is provided for Chile.

# 1. INTRODUCTION

The relationship between economies of agglomeration and local entrepreneurship rates is not straightforward. Although usually documented to be positive (Audretsch and Fritsch, 1994; Van Soest et al., 2006; Sato et al., 2012), some empirical evidence suggests that the relationship could be insignificant or perhaps even negative (Naudé et al., 2008; Behrens and Robert-Nicoud, 2015). This paper presents a model that addresses, within a New Economic Geography (NEG) framework, the sometimes seemingly counter-intuitive relationship between the spatial distributions of entrepreneurship and regional market potential (as proposed by Hanson, 2005).

Acknowlegments: This paper has greatly benefited from comments by JRS editor Professor Jens Suedekum and three anonymous reviewers to whom we are very grateful. The authors also want to thank María Sánchez-Vidal, Alexander Lembcke and participants at the 2014 INFER Workshop in Urban and Regional Economics held in Reus, Spain, for helpful comments and suggestions. The usual disclaimer applies. The National Commission of Scientific and Technological Research of Chile (CONICYT) supported this work under grant [FONDECYT 1130356].

Received: October 2015; revised: May 2016; accepted: August 2016.

Our model explicitly includes both push and pull mechanisms conditioning the spatial distribution of entrepreneurship. More specifically, we introduce occupational choice and heterogeneous managerial ability, two features of microeconomic theories of entrepreneurship (Parker, 2009), into the spatial Dixit-Stiglitz setting (Fujita et al., 1999). The resulting model predicts a positive partial effect of market potential, and a negative partial effect of domestic wages on the regional supply of entrepreneurs. But wages are also determined by the regional market potential, and so the model predicts, in the short-run equilibrium, offsetting effects of MP over the regional supply of entrepreneurs. The long-run equilibrium result, after full mobility of workers across regions and occupations, is that differences in entrepreneurship rates are given by differences in fixed costs across regions. We provide an empirical application of the model, by econometrically testing the offsetting effects of regional market potential on short-run changes in municipal entrepreneurship rates in Chile. The evidence supports the model's theoretical implications.

Our model introduces several innovations to existing NEG-based models of location and entrepreneurship. First, in contrast to Sato et al. (2012), we incorporate and test for interregional demand linkages as an explanation for the spatial variation in entrepreneurship, using a supply equation relating regional entrepreneurship rates and the Helpman-Hanson's market potential function (Helpman, 1998; Hanson, 2005). Second, in contrast to what are known as "footloose entrepreneurs" models of agglomeration (e.g., Forslid and Ottaviano, 2003), where there is an exogenous stock of entrepreneurs choosing their location, our model focuses on the effects of regional MP in the individual's decision to become an entrepreneur. Third, unlike Sato et al. (2012), our model endogenizes both business profits and wages, both conditioning the selection into entrepreneurship. The model yields testable parameter restrictions reflecting the trade-offs between regional incentives and the opportunity costs of entrepreneurship. Finally, it incorporates entrepreneurs' heterogeneity, leading to the sensible implication that, ceteris paribus, less efficient firms can succeed in areas of higher market potential and that average firm size should be greater in such places. Moreover, the result that long-run differences in local entrepreneurship rates are given by differences in fixed costs is consistent with location drivers in spatial equilibrium (Roback, 1982; Wu and Gopinath, 2008) and urban economics (Glaeser, Rosenthal, and Strange, 2010; Glaeser, Kerr, and Ponzetto, 2010) models, as well as with empirical evidence in both developed (Glaeser, Kerr, and Ponzetto, 2010) and developing countries (Naudé et al., 2008; Modrego et al., 2014). These equilibrium results could differ, however, both in the short run and also in conditions where long-run interregional migration flows might be restricted.

Chile serves as a good case for testing the implications of the model. First, the Global Entrepreneurship Monitor (GEM) Project—a large, long-term cross-country study of the drivers of entrepreneurship (Reynolds et al., 2005; Bosma and Levie, 2010)—portrays the country as one of the most entrepreneurial countries in the world, notably out-performing other countries at similar stages of development (Figure 1). While the literature emphasize the importance of a greater market size for entrepreneurship (Murphy et al., 1990; Reynolds et al., 2012), Figure 1 shows that Chile's entrepreneurship rate is well above that of countries with a much higher per capita GDP (United States, for example).

Second, since the mid-2000s, enhancing entrepreneurship has become a key component of Chilean economic policy, and more than 100 entrepreneurship support programs have been established in the country (Romaní et al. 2009). However, we know little about the spatial dimension of entrepreneurship in the country. Given its fundamental importance for regional economic growth and development (Acs and Armington, 2004; Audretsch and Keilbach, 2004), understanding more precisely the drivers of spatial differences in entrepreneurial activity is of critical importance for both academic and policy purposes.



*Note:* GDP per capita in constant U.S.\$ of 2005. Source: Own elaboration based on data from the GEM project and the World Bank, accessed through World Statistics (http://world-statistics.org).

FIGURE 1: Entrepreneurial Activity and National Development (2012 or Closest).

Third, what we know is that regardless of how it is measured, entrepreneurship in Chile, as in most countries, is very unevenly distributed over space. Modrego et al. (2014) report differences in the number of firms per capita as large as 14 times between municipalities, and differences in annual business start-up rates varying from less than 1–33 new activities per 1,000 inhabitants (Modrego et al. 2015). Figures 2(a) and (b) show, as of the early 2000s, the market potential of Chilean municipalities and rates of nonfarm self-employment respectively. While there are high entrepreneurship rates in agglomerated areas, such as the Santiago Metropolitan Region, high entrepreneurship rates are also evident in many rural, remote communities, such as northern Araucanía (one of the poorest areas in the country). Apparently, the relationship between agglomeration and local entrepreneurship rates in Chile is not straightforward as well.

The following section sets out an NEG-based framework explaining the relationship between location and regional entrepreneurship rates. Section 3 presents an econometric application of the model in the previous section. Section 4 presents the data used for testing the model at the level of Chilean municipalities. Section 5 discusses the results and section 6 concludes and discusses some pending lines for future research.

#### 2. A SPATIALLY-EXPLICIT OCCUPATIONAL CHOICE MODEL

A main feature of NEG models is providing general equilibrium solutions for the spatial distribution of economic activity, as a balance between agglomeration forces arising from increasing returns and transport costs and dispersion forces stemming from an immobile demand source (Krugman, 1991) or a nontradable sector (Helpman, 1998). Following mainstream occupational choice theories (Parker, 2009), we incorporate here another trade-off, that originating in the discrete choice each individual faces regarding becoming an entrepreneur or an employee.



*Note:* In both panels, Northern Chile at the left, the Center–South area in the center and the South end at the right. Source: Panel (a) Modrego et al. (2014) and panel (b) 2002 National Population Census.

# FIGURE 2: Harris (1954) Market Potential and Nonfarm Self-employment Rates in Chilean Comunas (2002/2003).

We retain most of the assumptions in Helpman (1998). The economy comprises R regions and a mobile population of size  $L_r$  in each region r. Later we will introduce mobility and individuals' sorting across regions based on real earnings equalization (Fujita et al., 1999; Hanson, 2005).

There are two sectors in the economy: (i) a monopolistically competitive sector with increasing returns and costly trade across regions ("manufactures") and (ii) a homogeneous good not traded across regions (H).<sup>1</sup> In each region, this nontradable good is supplied in fixed stock  $H_r$ , and following Helpman (1998) and Hanson (2005) we further assume that total stock of the nontradable good is equally owned by individuals in the economy, regardless of regional location. Individuals supply inelastically one unit of labor, but they choose between paid employment (l) and entrepreneurship (e). As in Lucas (1978), Evans and Jovanovic (1989), Parker (2005) and others, each individual is endowed with an intrinsic level of managerial ability  $(\theta)$ , drawn from a short-run fixed (continuously differentiable) regional distribution:  $f^r(\theta) \rightarrow [A, +\infty]$ , with A > 0 being the minimum level of ability in the population.

On the demand side, we introduce no innovations to the standard multi-region coreperiphery model (Fujita et al., 1999). Individuals have Cobb-Douglas preferences over consumption of a composite of manufactures (m) and housing (h), constant across regions:  $U = m^{\mu} h^{1-\mu}$ , with the parameter  $\mu$  indicating the budget share of manufactures. The composite variable m is a constant-elasticity-of substitution (CES) specification:  $m = [\int_0^n m_i^{\rho} di]^{1/\rho}$ , with  $m_i$  being consumption of each variety i and  $\sigma \equiv 1/(1-\rho)$  is the elasticity of substitution between varieties.

Each manufactured variety *i* is produced in region *r* at price  $p_{ir}$  and is sold across regions with trading costs. We assume iceberg transport costs, such that its price in external market *s* is (Hanson, 2005):  $p_{is} = p_{ir}e^{\kappa d_{rs}}$ , with *d* being the distance between markets *r* and *s* and  $\kappa$  a parameter measuring unit transport costs. Given assumptions, the demand for the *i*th differentiated variety arising from consumer's utility maximization is expressed in terms of the Market Potential (MP) function (see Fujita et al., 1999; Hanson, 2005):

(1) 
$$q_{i_r}^d = \mu p_{i_r}^{-\sigma} M P_r \equiv \mu p_{i_r}^{-\sigma} \sum_{s=1}^R Y_s e^{\kappa(1-\sigma)d_{rs}} G_s^{(\sigma-1)}.$$

with Y being total regional income and  $G_s$  the CES price index for region s. MP is the regional market potential, a synthetic measure of the size of surrounding markets in terms of incomes adjusted for purchasing power, negatively weighted by distance-dependent transport costs (Hanson, 2005).

On the supply side, each entrepreneur (firm) produces a different variety and the production technology entails the use of a variable input of labor, here normalized to one, plus an input of labor fixed to production levels (the source of increasing returns). Labor is paid at a regional wage,  $w_r$ , which also represents the opportunity cost of the entrepreneur, who can always earn this wage if he decides to be an employee. To introduce entrepreneurs' heterogeneity in the production technology, we borrow from Glaeser, Kerr, and Ponzetto (2010), the idea of a managerial overhead cost  $\tau > 0$ , here assumed to be decreasing in the individual's managerial ability  $\theta_{ir} (\geq A)$ . Accordingly, we specify the following cost function:

(2) 
$$CT_{ir} = w_r \left(q_{ir} + k_{ir}\right) = w_r \left(q_{ir} + \tau \theta_{ir}^{-\gamma}\right),$$

with  $k_{ir}$  being the firm-specific fixed input of labor. Although these costs are fixed with respect to production levels for a given firm, the (constant across firms and regions) parameter  $\gamma > 0$  represents an elasticity of fixed costs with respect to the entrepreneurs' "managerial ability," that is, it is a managerial efficiency parameter. Despite the extreme

<sup>&</sup>lt;sup>1</sup>As a reviewer correctly points out, in the short run this nontradable sector is superfluous, but it is useful to motivate differences in the cost of living in the empirical section and also we will discuss it in the context of long-run equilibrium analysis. We thank the reviewer for this observation.

<sup>© 2016</sup> Wiley Periodicals, Inc.

simplification of the entrepreneurial function, this representation of the production technology in a market structure of potentially-unlimitedly differentiated varieties is consistent with a view of the (able) entrepreneur as the individual (more) capable of recombining resources in more efficient ways to innovate and create value (Carree and Thurik, 2003).<sup>2</sup>

Given this cost structure, the profit function of the individual entrepreneur i located in region r is:

(3) 
$$\pi_{ir} = (p_{ir} - w_r) q_{ir} - w_r \tau \theta_{ir}^{-\gamma}$$

The monopolistic entrepreneur in the differentiated sector sets prices such that marginal benefit equals marginal cost; here, this yields the constant mark-up rule. Given marginal costs independence of individual ability, the price is the same for each variety produced in the region r as in the standard NEG core-periphery model:

$$(4) p_r = \frac{\sigma w_r}{\sigma - 1}$$

with  $\sigma$  also being the price elasticity of the demand for any variety produced in region r, with  $\sigma > 1$  due to the maximizing monopolist's operating in the elastic part of the demand curve.

Embedding the pricing rule (4) and the demand equation (1) into the profit function (3) this latter can be rewritten as:

(5) 
$$\pi_{ir} = \phi M P_r w_r^{1-\sigma} - w_r \tau \theta_{ir}^{-\gamma},$$

with  $\phi = \mu \sigma^{-\sigma} (\sigma - 1)^{\sigma - 1}$ .

#### Characterizing the Equilibrium in the Short Run

We wish to examine equilibrium conditions when factors within a region can move freely; in particular, we define the short-run equilibrium as wages and profits that make worker indifferent to becoming entrepreneurs. Later we will address the long-run equilibrium, which we define as labor flows across regions based on differential real earnings (e.g., Fujita et al., 1999; Hanson, 2005).

Most previous NEG-based models incorporating entrepreneurship rely on the zeroprofits entry condition to derive the equilibrium relationships. However, as pointed by Glaeser, Rosenthal, and Strange (2010, p. 4), the zero-profits equilibrium "takes entrepreneurship out of the model," by implicitly assuming a horizontal supply of entrepreneurs across space. To incorporate the individual decision of starting a business in this spatial economic model, we follow an occupational choice setting. As in Lucas (1978), Evans and Jovanovic (1989), Parker (2005) and others, workers choose entrepreneurship as long as the returns to self-employment are at least as high as those of hired labor. The short-run (long-run) free-entry occupational choice equilibrium condition for all ability levels choosing entrepreneurship is then:

(6) 
$$\pi_{ir} \geq w_r \quad \forall r.$$

<sup>&</sup>lt;sup>2</sup>One could think of alternative ways of incorporating differences in managerial ability. For instance, occupational choice models usually assume that greater ability translates into greater productivity via shifts outward in the production function (e.g., Lucas, 1978; Evans and Jovanovic, 1989). Alternatively, one could assume that managerial ability also entails marginal cost-efficiency. Both approaches add algebraic complications, but are inessential to the main thrust of the model, which is that endogenous entrepreneurs' opportunity costs (wages) offset pro-entrepreneurship market-size effects.

<sup>© 2016</sup> Wiley Periodicals, Inc.



FIGURE 3: Partial Effects of Increases in Market Potential and Domestic Wages on the Short-Run Regional Equilibrium Level of Managerial Ability.

At the ability level where entrepreneurial income is equal to wages produces a *regional marginal ability function*  $(z_r)$  defining the point at which the marginal entrepreneur located in region r breaks even, that is, the point at which incomes from the entrepreneurial undertaking equals the total costs (including production and the opportunity cost of not being an employee). This function is decreasing in the regional MP and increasing in domestic wages. By inserting equation (5) into (6) at the equilibrium level of ability yields

(7) 
$$z_r = \tau^{1/\gamma} \left( \phi M P_r w_r^{-\sigma} - 1 \right)^{-1/\gamma}.$$

Equation (7) states that, everything else constant, in areas of higher market potential, a lower level of managerial ability is needed to succeed in business, since interregional demand linkages provide opportunities for marginal (less able) entrepreneurs to make profits. The effect of an increase in the market potential is illustrated in Figure 3 as the shift to the left in the cut-off level of ability (from z to z'). The model thus retains the idea of larger demand as a favorable condition for entrepreneurship (Sato et al., 2012; Modrego et al., 2014). But at the same time, it casts a positive partial effect of domestic wages over z. Since the individual can access this wage (which is also what the entrepreneur must pay his employees) by choosing being an employee, when wages are higher, less able potential entrepreneurs will rationally opt to be paid employees (Lucas, 1978; Parker, 2005). This can be seen in Figure 3 as the displacement of the cut-off level of ability to the right, from point z to z". Given that  $z_r$  is nonnegative, then in equilibrium wages would be such that:  $(\phi MP_r)^{1/\sigma} \ge w_r$ , and so assuring an interior solution with both entrepreneurs and workers in the region.<sup>3</sup>

In this model, relationship (7) establishes a subtle but crucial difference from Sato et al. (2012), where regional variations in the opportunity cost of entrepreneurship (wages) are taken as exogenously determined. In this present treatment, we instead make the entrepreneur's opportunity cost endogenous, following one of the main implications of the NEG core-periphery model, which is the positive dependence of regional wages on regional market potential (Fujita et al., 1999; Hanson, 2005).

<sup>&</sup>lt;sup>3</sup>We thank an anonymous reviewer for motivating this characterization of the equilibrium solution.

<sup>© 2016</sup> Wiley Periodicals, Inc.

Our strategy is to develop the labor market equilibrium to check whether this NEG's core relationship is affected by the introduction of occupational choice elements in the model. We also assess the net effect of a shock in the MP on the regional supply of entrepreneurs. These net effects are both direct through business profits and indirect through equilibrium wages.

The short-run regional supply of entrepreneurs is the number of people with a level of ability above the cut-off point:

(8) 
$$e_r = L_r \int_{z_r}^{+\infty} f^r(\theta) d\theta = L_r \left[ 1 - F^r(z_r) \right]$$

Analogously the regional short-run supply of employees  $(l^s_r)$  is

$$l^{s}{}_{r} = L_{r}F^{r}(z_{r}).$$

Given assumed technology, the firm-level demand for labor  $(l^{d}_{ir})$  is

$$l^{d}_{ir} = q_{ir} + \tau \theta_{ir}^{-\gamma} = \omega M P_r w_r^{-\sigma} + \tau \theta_{ir}^{-\gamma},$$

with  $\omega = \phi/(\sigma - 1)$ .<sup>4</sup>

This means, given the way the managerial ability enters the cost function, that less able entrepreneurs in the region will demand more labor per output, generating higher unit costs. In the absence of agglomeration externalities on production (as in Sato et al., 2012), this means that *keeping everything else constant*, regions with higher market potential will have more firms, of larger average size and lower average labor productivity.

In turn, the regional demand for labor is given by (see Lucas, 1978):

(10) 
$$l_r^d = L_r \int_{z_r}^{+\infty} l_{ir}^d(\theta) dF^r = L_r \left( 1 - F^r(z_r) \right) \left[ \omega M P_r w_r^{-\sigma} + \overline{k}^r(z_r) \right].$$

This is equivalent to the total number of entrepreneurs (firms) in equation (8) (the term outside brackets), multiplied by the average labor input per firm (the term in brackets). The latter is comprised of the (constant across firms in the region) variable labor input (first term in brackets) and the average fixed labor input per firm in the region  $(\overline{k}^r(z_r))$ ). This regional labor demand is increasing in the MP and decreasing in domestic wages (as are both the demand faced by each firm and the number of firms).

Equalization of the regional labor supply (9) and the regional demand for workers (10) defines the equilibrium regional wages as a nonlinear function of the MP:

(11) 
$$w_r = \left[ \left( \frac{F_r(z_r)}{(1 - F_r(z_r))} - \overline{k}^r(z_r) \right) \frac{1}{\omega M P_r} \right]^{-1/\sigma}.$$

An implication of equation (11) is that *ceteris paribus*, a higher share of entrepreneurs translates into higher wages, because more entrepreneurship increases labor demand while reducing salaried-labor supply. Equations (7) and (11) solve the model for the regional equilibrium number of entrepreneurs (and employees) and regional wages as a function of the MP. Two main results, which are also testable hypotheses of the model, that arise from the short-run equilibrium conditions are:

**Hypothesis 1**: Regional wages are increasing in the market potential. More specifically, as in the standard NEG core-periphery setting, the elasticity of wages with respect to changes in the MP is the inverse of the elasticity of substitution.

<sup>&</sup>lt;sup>4</sup>This result is an application of Hotelling's lemma to the profit function as given in equation (5).

<sup>© 2016</sup> Wiley Periodicals, Inc.



*Note:* Greater market potential shifts labor demand rightward, from  $l^s_r$  to  $l^s_r$ ', raising wages from  $w^*_r$  to  $w^*_r$ ' (from point A to point B). Increased wages draws marginal entrepreneurs to switch back into paid employment, thus increasing the supply of labor and decreasing demand (from point B to point C). The net effect of these changes is a new equilibrium with the same share of entrepreneurs and paid workers but with higher wages, as point C.

# FIGURE 4: The Short-Run Regional Labor Market Equilibrium—Effects of an Increase in the Market Potential.

**Hypothesis 2**: In the short run, before wage differentials trigger immigration, the equilibrium share of entrepreneurs is invariant to shocks in the MP, as the higher profit-effect is completely offset by the opportunity-cost effect.

We leave to appendix 1 and 2 the formal proof of hypotheses 1 and 2, as an exercise in comparative statics motivated by a marginal change in the MP. Still, one simple way of analytically verifying the second hypothesis is to apply equation (11) defining the local labor market equilibrium wage to equation (7) defining the regional marginal ability function to note that the indifference level of ability is determined by the regional distribution of abilities and not reference to market potential or other variables:<sup>5</sup>

$$z_r = au^{1/\gamma} igg[ (\sigma-1) igg( rac{F^r(z_r)}{1-F^r(z_r)} - \overline{k}^r(z_r) igg) - 1 igg]^{-1/\gamma}$$

To develop the intuition of these results here, Figure 4 depicts the short-run (before wage-driven immigration) adjustments in the labor market triggered by a positive shock in regional MP, such as the building of a new road or a new (spatial targeted or not) public transfer program. The initial effect of an increased MP is a decrease in the regional break-even level of ability. This provokes a shift to the left of the supply of workers, as some individuals previously below the cut-off level of ability switch into entrepreneurship. Incumbent entrepreneurs now individually demanding more labor and new entrepreneurs demanding increasingly more labor shift the demand curve to the right, forcing wages to go

<sup>&</sup>lt;sup>5</sup>We would like to thank a *JRS* reviewer for suggesting this interpretation.

<sup>© 2016</sup> Wiley Periodicals, Inc.

up to restore the equilibrium (from point A to point B). Wages will increase to the original break-even level of ability, making marginal entrepreneurs switch into paid employment. This shifts the supply for labor to the right and the demand to the left. As demonstrated in the appendix 1, the net effect of these changes is a new equilibrium with the same share of entrepreneurs and employees but with higher wages, as point C in the figure. In fact, the second hypothesis also holds in the long run, as will be demonstrated below.

In this NEG-based model with only one production factor, the increase in wages is the only way the surplus generated by the increase in MP is shared between entrepreneurs and employees. Additional rent is bid into both the entrepreneurial profits and into the wages of workers. The relative gains of the two groups depend on the size of the groups, but as the appendix 2 demonstrates, all participants regardless of ability ( $\theta$ ) gain proportionally the same amount ( $1/\sigma$ ). The greater is the elasticity of demand, the smaller would be the potential mark-up of prices over marginal costs and smaller would be the total rents available to be shared between both entrepreneurs and employees.

#### Long-Run Equilibrium Results When Labor Is Mobile Across Regions<sup>6</sup>

In the long run, both entrepreneurs and employees can migrate across regions. To characterize the long-run equilibrium, we examine the three short-run endogenous variables of interest in the previous section: entrepreneur profits at every ability level (equation 5), the regional marginal ability function (equation (7)), and regional nominal wages, which equilibrate the local labor market (equation (11)). These are functions of a long-run endogenous variable: the Market Potential function (defined in equation (1)). Market potential is, in turned, defined by the regional price index (see Fujita et al., 1999; Hanson, 2005):

(12) 
$$G_r = \left[\sum_{s=1}^R n_s \left(\frac{\sigma w_s}{\sigma - 1} e^{\kappa d_{rs}}\right)^{1 - \sigma}\right]^{\frac{1}{1 - \sigma}},$$

with  $n_s$  being the number of varieties of manufactures, which is equal to the number of firms, and to the number of entrepreneurs in regional market *s*:

$$n_s = e_s = L_s \int_{z_s}^{\infty} f^s(\theta) d\theta$$

An additional (short-run and long-run) equilibrium condition is the clearing of the nontradable market with inelastic supply (total payments equating total expenditures) (Hanson, 2005):

(13) 
$$p_r^h H_r = (1-\mu) Y_r, \ \forall r,$$

such that total regional incomes are defined as

(14) 
$$Y_r = L_r \left( F^r(z_r) w_r + \int_{z_r}^{+\infty} f^r(\theta) \left( \phi M P_r w_r^{1-\sigma} - w_r \tau \theta_i r^{-\gamma} \right) d\theta + \frac{1}{L} \sum_{s=1}^R p_s^h H_s \right).$$

The first term in the parenthesis of equation (14) (when multiplied by  $L_r$ ) represents the wages times the number of employees; the second is the entrepreneurs' mean earnings

<sup>&</sup>lt;sup>6</sup>This subsection was absent in a previous version of the paper and it was motivated by the comments of three *JRS* reviewers to whom we are extremely grateful.

<sup>© 2016</sup> Wiley Periodicals, Inc.

times the number of entrepreneurs, and the third is the income from the nontradable sector.

Some long-run equilibrium results fall out of assuming an interregional labor market equilibrium with costless migration, which means that no individual, regardless of the ability level, would have an incentive to move between regions and/or occupations. Following Hanson's (2005) treatment of labor market equilibria with costless migration, real wages depend on nominal wages deflated by a regional price index (defined by the consumers' expenditure function) and equal across regions:<sup>7</sup>

(15) 
$$\frac{w_r}{\left(p_r^h\right)^{1-\mu}G_r^{\mu}} = \frac{w_s}{\left(p_s^h\right)^{1-\mu}G_s^{\mu}}, \ \forall r, s.$$

Similarly, in this present model, the real profits of entrepreneurs (individuals at or above the cut-off level of ability) must also be equal across regions at every ability level, otherwise entrepreneurs would migrate, either to be entrepreneurs in another region or to be an employee:

(16) 
$$\frac{\pi_{ir}}{\left(p_r^h\right)^{1-\mu}G_r^{\mu}} = \frac{\pi_{is}}{\left(p_s^h\right)^{1-\mu}G_s^{\mu}} \quad \forall \ \theta_i \ge z_t \text{ and } \forall \ t = r, \ s.$$

Note that free-entry occupational choice equilibrium (equation (6)) means also that real earnings are equal across regions and occupations at the indifference ability level of the marginal entrepreneur:

(17) 
$$\frac{\pi_r^z}{\left(p_r^h\right)^{1-\mu}G_r^{\mu}} = \frac{w_r}{\left(p_r^h\right)^{1-\mu}G_r^{\mu}} = \frac{w_s}{\left(p_s^h\right)^{1-\mu}G_s^{\mu}} = \frac{\pi_s^z}{\left(p_r^h\right)^{1-\mu}G_r^{\mu}}, \ \forall r, s.$$

Noting that profits at all ability levels are proportional to wages:

$$\pi_{ir} = \phi M P_r w_r^{1-\sigma} - w_r \tau \theta_i^{-\gamma} = w_r \cdot (\phi M P_r w_r^{-\sigma} - \tau \theta_i^{-\gamma}),$$

condition (16) combined with the symmetry of production technology across regions entails that, in the long run, all entrepreneurs of the same ability level  $\theta$  earn the same premium above wages:

$$\phi MP_r w_r^{-\sigma} - \tau \theta^{-\gamma} = \phi MP_s w_s^{-\sigma} - \tau \theta^{-\gamma} \quad \Rightarrow MP_r w_r^{-\sigma} = MP_s w_s^{-\sigma}.$$

And in particular, the equilibrium ability level is the same for all regions:

(18) 
$$z^r = z^s = \overline{z}, \ \forall r, s.$$

This latter result suggests another observable consequence of the theory, at least in the long run: in equilibrium, the relative *nominal* wages between two regions are proportional to the ratio of regional market potentials, as also is the cost of living ratio:

(19) 
$$\frac{w_r}{w_s} = \left(\frac{MP_r}{MP_s}\right)^{1/\sigma} = \left(\frac{p_r^h}{p_s^h}\right)^{1-\mu} \left(\frac{G_r}{G_s}\right)^{\mu}.$$

<sup>&</sup>lt;sup>7</sup>As remarked by one *JRS* reviewer, recent NEG models (e.g., Südekum, 2008) emphasizing the role of amenities, propose a spatial equilibrium sorting based on the equalization of indirect utilities. Because our focus is placed on the role of NEG's market potential function in conditioning the regional supply of entrepreneurs, we resort to the standard real wage equalization mechanism followed by Fujita et al. (1999) and Hanson (2005), among others. In absence of amenities, the principle of deflating nominal wages is exactly the same whether one equates real wages or indirect utilities. We thank the reviewer for bringing up this point.

Relationship (19) reflects the standard type of interregional, economic-geography equilibrium model with the result that firms must compensate workers for living in a costly (or unpleasant/congested) place (Roback, 1982; Wu and Gopinath, 2008). Firms do so by offering higher wages, which is feasible since firms enjoy some location advantage, in this case greater profits due to market proximity (a greater market potential).

Finally, given that the marginal abilities are equal across regions one notes that in this model, long-run differences in entrepreneurship rates are, given the distributions of abilities in regions, independent of all other variables:

$$(20) \quad \frac{F^{*}(\bar{z})}{1 - F^{*}(\bar{z})} - \frac{F^{*}(\bar{z})}{1 - F^{*}(\bar{z})} = \overline{k}^{r}(\bar{z}) - \overline{k}^{*}(\bar{z}) = \int_{\bar{z}}^{\infty} \tau \theta^{-\gamma} f^{r}(\theta) d\theta - \int_{\bar{z}}^{\infty} \tau \theta^{-\gamma} f^{*}(\theta) d\theta$$

This result can be verified by inserting equation (11) into (7) and using result (17). Long-run equilibrium differences in entrepreneurship rates depend in differences in average fixed costs between regions. In other words, regions with higher average fixed costs of doing business would have lower entrepreneurship rates, a result that is also present in urban economic models of entrepreneurship (Glaeser, Kerr, and Ponzetto, 2010; Glaeser, Rosenthal, and Strange, 2010). For instance, if the regional density distributions of entrepreneurial ability are equal, then average fixed costs and entrepreneurship rates will also be equal. This is not say that the distributions of abilities by region are themselves not functions of the more basic NEG parameters, such as the stock of the nontradeable good,  $H_r$ , or the transport costs between regions,  $\kappa d_{rs}$ . On the contrary, such distributions depend on the nature and patterns of migration behavior in the context of the geographical features relating to the supply of housing and infrastructure, all of which themselves shape market potential. Based on this model, there are no *a priori* conclusions as to what these patterns might look like in the long run.

As noted above, a positive shock to MP in a region delivers proportional gains across all ability levels, which in the long run would induce immigration from other (less fortunate) regions. If one were to assume that migration costs are independent of the entrepreneurial ability level, a reasonable assumption in the light of Lazear's "jack-of-alltrades" hypothesis (Lazear, 2004, 2005), this would imply immigration from all talent levels.<sup>8</sup> In effect, migration from other regions would be a *pure scale effect*, as if peeling a layer from the talent distribution of the rest of the world and laying it on top of that of the region enjoying the increase in market potential.<sup>9</sup> If, on the contrary, one expects a negative correlation between entrepreneurial ability and the cost of migration, the result would be agglomeration with increased entrepreneurship rates (a dynamic that is inconsistent with stylized facts indicating a negative correlation between urban density and self-employment rates, see Behrens and Robert-Nicoud, 2015).

The exactly compensating profit and opportunity cost effects driven by changes in market potential may seem, at first glance, an intuitively puzzling result, given the

<sup>&</sup>lt;sup>8</sup>Although it is usually assumed that entrepreneurs are more educated people, recent profiles show that self-employed and employees in Chile are not very different in terms of formal schooling (10.1 and 11.5 years respectively) (Modrego, Paredes, and Romaní, 2015).

<sup>&</sup>lt;sup>9</sup>Given an even stronger assumption of similar initial ability distributions across regions, and the similar proportional gains across all talent levels that would induce immigration from elsewhere, the final equilibrium distribution of ability in the beneficiary region would not change. The absolute number of entrepreneurs and employees would increase, driving down local entrepreneurial real rents and wages; and the absolute number of entrepreneurs and employees elsewhere would decline, driving up real rents and wages. A final equilibrium, long-run result would be one region with a larger population but with the same shares of workers and entrepreneurs, and a sharing across all regions of the gains due to the region's enhanced market potential.

theories and empirical evidence supporting the positive relationship between market size or market potential and entrepreneurship (Audretsch and Fritsch, 1994; Sato et al., 2012) and business density (Modrego et al., 2014). However, as remarked by Behrens and Robert-Nicoud (2015), whether the share of entrepreneurs increases or decreases with market size or density strongly depends on the modelling choices and empirical evidence is ultimately inconclusive. In this present application, the result directly stems from the model's structure: the constant mark-up rule, the production technology with only fixed costs depending on the managerial ability, and the free-entry equilibrium condition of entrepreneurs given by the equalization of business profits and wages. Combined, these conditions yield a proportional relationship between the opposing incentives conditioning the selection into entrepreneurship, which in turn leads to these symmetric shifts in the labor market and to long-run differences independent of regions' MP.<sup>10</sup> These predictions can be interpreted as consistent with some empirical evidence from developed economies discarding the hypothesis of places with abnormal business returns as a main explanation for the spatial variation of entrepreneurship (Glaeser, Kerr, and Ponzetto, 2010). As shown in the next section, the compensating rents and wage effects fit well with the case of municipal entrepreneurship rates in Chile.

#### 3. AN EMPIRICAL APPLICATION

One important implication of the model developed above is the offsetting effect of location in relation to market with regard to determining the short-run equilibrium supply of entrepreneurs in a region. In this section, we present an econometric strategy that uses short-term differences in municipal entrepreneurship rates in Chile to test the short-run equilibrium hypotheses presented in the previous section.

We start from the regional supply of entrepreneurs (equation (8)). To arrive at a specific functional form, we follow Lucas (1978) and assume a Pareto distribution function for the entrepreneurial ability:  $F(\theta) = 1 - A^{\eta}\theta^{-\eta}$ , with  $\eta > 0$  being a parameter that here measures the elasticity of the supply of entrepreneurs to the managerial ability.

Expressing the equilibrium supply of entrepreneurs (8) in rates and totally differentiating yields

(21) 
$$d\ln\left(\frac{e}{L}\right)_r = -\eta \cdot d\ln z_r.$$

Taking the total differential of the regional marginal ability function (equation (A.1) in the appendix 1) and rearranging terms:

<sup>&</sup>lt;sup>10</sup>There are ways to modify this sharp result of the model. One is to assume an additive utility function that includes a subjective element of personal satisfaction of being self-employed, as in Blanchflower and Oswald (1998). In such case, the individual equates the sum of pecuniary and nonpecuniary utility of entrepreneurship with the wage income utility of being an employee. In equilibrium, marginal entrepreneurs will end up earning less than the regional wage and the opportunity-cost effect will more than compensate the entrepreneurial-utility effect, leading to lower entrepreneurship rates in presence of an increase in the MP. While appealing, such an approach would create an identification problem in the empirical strategy we devise to test the trade-offs driven by location over entrepreneurship, because it introduces an additional unobservable parameter. Another option is to include a spatially-variant start-up cost, which would lead to the opposite equilibrium result. We lack of an appropriate proxy in the empirical analysis.

which is a good approximation provided  $MP_r$  is sufficiently large relative to  $w_r$ .<sup>11</sup>

Combining equations (21) and (22):

(23) 
$$d\ln\left(\frac{e}{L}\right)_r \approx \frac{\eta}{\gamma} d\ln\left(MP\right)_r - \frac{\eta\sigma}{\gamma} d\ln\left(w\right)_r.$$

Equation (23) reflects the equilibrium relationship as derived from the theoretical model. However, in practice observed entrepreneurship rates likely deviate from such a state, so we propose the following dynamic linear reduced form:

(24) 
$$\ln\left(\frac{e}{L}\right)_{rt} = \beta_0 + \beta_1 \ln\left(\frac{e}{L}\right)_{rt-1} + \beta_2 \ln\left(MP\right)_{rt} + \beta_3 \ln\left(w\right)_{rt} + \lambda \ln\left(\mathbf{x}\right)_{rt} + \nu_r + \delta d_t + \varepsilon_{rt}.$$

In equation (24),  $\beta_1$  captures path dependencies in regional entrepreneurship dynamics (Fritsch and Mueller, 2007; de Blasio and Nuzzo, 2010; Andersson and Koster, 2011).  $\nu_r$  accounts for time-invariant features of regional economies potentially affecting the supply of entrepreneurs, such as institutions (Saxenian, 1994; Boettke and Coyne, 2003) or regional entrepreneurial culture (Davidson and Wiklund, 1997). In terms of the model developed above, these fixed effects are accounting for region-specific parameters shaping the different regional ability distribution functions ( $f^r(\theta)$ ). The vector **x** contains other economic and demographic variables suggested in the literature as potentially conditioning the supply of entrepreneurs (Reynolds et al., 1995; Blanchflower and Oswald, 1998), and are discussed in the data section. Term  $d_t$  are time dummies capturing transitory shocks common to all regions.

Time-differencing (24) eliminates the regional fixed effect, such that the final estimation equation is

$$(25) \Delta \ln \left(\frac{e}{L}\right)_{rt} = \beta_1 \Delta \ln \left(\frac{e}{L}\right)_{rt-1} + \beta_2 \Delta \ln (MP)_{rt} + \Delta \beta_3 \ln (w)_{rt} + \lambda \Delta \ln (\mathbf{x})_{rt} + \delta d_t + \Delta \varepsilon_{rt}.$$

Equation (25) is the empirical counterpart of our comparative statics exercise in Figure 4 and appendix 1. It models the process of net entry into entrepreneurship as a response to changes in the regional market potential and regional wages.

Our main tests of hypothesis are the following:

(26a) 
$$\beta_2 > 0, \ \beta_3 < 0.$$

A more stringent implication arising from relationship (23) is

$$(26b) -\frac{\beta_3}{\beta_2} = \sigma > 1$$

consistent with the monopolist's maximizing behavior. Previous estimates of parameter  $\sigma$  for Chile being around 1.5–1.8 (Modrego et al., 2014) provide a benchmark for further validation of the model.

To compute regional market potential, we follow Hanson (2005) and make use of the equilibrium sorting of workers, given by equalization of real earnings across regions (equation (16)). Solving for  $G_s$  in (15) and raising the resulting expression to ( $\sigma - 1$ ), yields (see Hanson, 2005):

(27) 
$$G_s^{\sigma-1} = C w_s^{\frac{\sigma-1}{\mu}} \left( p_s^h \right)^{\frac{(\sigma-1)(\mu-1)}{\mu}}$$

For this empirical application, we follow the convention of conceptualizing the non-tradable sector (*H*) as housing services, such that  $p_s^h$  indicates housing prices. As discussed

<sup>11</sup>In our dataset  $\frac{\phi M P_r w_r^{-\sigma}}{\phi M P_r w_r^{-\sigma} - 1}$  ranges from 1 to 1.008.

<sup>© 2016</sup> Wiley Periodicals, Inc.

in Hanson (2005), C is a function of constant terms across regions and it equals the lefthand side of Equation (15). Replacing (27) in (1) and omitting the constant C yields the following market potential function used in the empirical application:

(28) 
$$MP_{r} = \sum_{s=1}^{R} Y_{s} e^{\kappa(1-\sigma)d_{rs}} w_{s}^{\frac{\sigma-1}{\mu}} \left( p_{s}^{h} \right)^{\frac{(\sigma-1)(\mu-1)}{\mu}}$$

Note that our main theoretical restriction entails the estimation of the elasticity-ofsubstitution parameter,  $\sigma$ , which is at the same time part of the market potential function (28), one of our two principal covariates. Instead of relying on nonlinear techniques that may carry some problems in the context of instrumental variables estimation (see Mion, 2004), we follow Fingleton (2006), Fingleton and Fisher (2010) and others in taking previous estimates of structural NEG parameters when computing the MP function in each region. We choose the following values estimated for Chile by Modrego et al. (2014) (average of columns 1–4 in Table 2):  $\mu = 0.87$ ,  $\kappa = 40$  and  $\sigma = 1.6$ . If the proposed theory fits the Chilean data, one should expect the term  $-\frac{\beta_3}{\beta_2}$  to not be different (statistically speaking) to this value of  $\sigma$  assumed for MP's calculation. Yet, since this *a prori* value chosen for  $\sigma$  may be considered small given other estimates reported in the literature (Mion, 2004; Hanson, 2005; Fingleton, 2006; Fingleton et al., 2010), we perform a robustness check by recomputing the MP variable for different elasticities of substitution, and then checking if relationship (26b) is still verified with these new MP's.

Given the endogeneity of our two main regressors (MP and wages) and due to the absence of good external instruments, model (25) is estimated by means of the generalized method of moments (GMM) estimator proposed by Arellano and Bond (1991) (A-B), implemented in Stata by Roodman (2009). This is currently a standard econometric technique for estimating dynamic panel models, and a powerful tool for reduced-form NEG equations with endogenous regressors (Mion, 2004). Despite that, there are still very few applications of the A-B method in empirical NEG studies. The estimates were carried using a two-step GMM estimator, which is robust in presence of heteroscedasticity and the Windmeijer (2005) finite sample correction was applied in order to avoid the well-known small-sample biases of coefficient's standard errors (Arellano and Bond, 1991).

#### 4. THE DATA

Our data source are ten rounds of the Chilean National Survey of Socioeconomic Characterization (CASEN) conducted in the years: 1990, 1992, 1994, 1996, 1998, 2000, 2003, 2006, 2009, and 2011.<sup>12</sup> The variables were collected at the level of *comunas* (municipalities), the lowest administrative unit in the Chilean administrative hierarchy. There were 342 comunas in 1990.<sup>13</sup> Of these 342 comunas, the variables were built for a group for which CASEN's design yields a representative sample in each round (called "self-represented comunas"). For such comunas, CASEN provides comuna-level weights for expanding the sample to the comuna's total population (see Mideplan, 2005), and such weights were used for expanding all variables built here. As such set of comunas varies in each round of the survey, we ended up with an unbalanced panel of 2,305 observations, which include observations for 331 of the 342 comunas in the country.

Our dependent variable (entrepreneurship rates in the theoretical model) is the proportion of self-employed plus employers to the working population, consistent with our

 $<sup>^{12} \</sup>rm There$  was also a round in 1987, which is currently unavailable in CASEN's website: http://observatorio.ministeriodesarrollosocial.gob.cl/casen\_obj.php

 $<sup>^{13}</sup>$ New comunas created during the period of analysis were treated as being part of the original comuna.

occupational choice framework and also a traditional metric in empirical analysis of entrepreneurial activity (see Parker, 2009). It excludes self-employed and employers in the natural resources (NNRR) primary sector (farmers, artisan fishermen), as entrepreneurship is usually associated with nonfarm proprietorship (e.g., Goetz and Rupasingha, 2009). Employees in the NNRR primary sector were also excluded as a model with a tradable industry under increasing returns and monopolistic competition is largely irrelevant in this context (Fingleton, 2006).

Following the standard practice, we calculate wages as the average income of the main occupation of employees in the comuna (Paredes, 2013). To avoid obvious aggregation biases, for all self-represented comunas in each CASEN round we calculate the municipal market potential in relation to a set of 51 comunas (the set s in equation 20) that are self-represented in all rounds in the 1990–2011 period. Although this way of constructing the MP variable comes at the cost of some measurement error, we believe that our empirical metric is still a good proxy of the true market potential. This set of 51 comunas encompasses most of the major local markets, including all the regional capital cities and most of the largest provincial capitals. Regarding the MP variables, regional incomes were taken as the sum of total (from all sources) incomes of all households in the comuna and housing prices were proxied by the average imputed housing rental value in the comuna. Monetary variables were expressed in monthly thousands Chilean pesos (base December 2009), deflated by the consumer price index reported by the Central Bank of Chile.<sup>14</sup> A matrix of distances between municipal capital cities was taken from the Ministry of Public Works and distances were expressed in thousands of kilometers.<sup>15</sup>

To test for the robustness of results in the base-specification, we include sociodemographic and economic variables shown as conditioning the regional supply of entrepreneurs. We first consider the share of municipal population above 24 years old with complete university (tertiary) studies, with the *a priori* expectation being an entrepreneurs' supply increasing, *ceteris paribus*, with greater levels of education, as more skilled individuals are more likely to succeed in self-employment (Robinson and Sexton, 1994; Parker, 2005). We also include the percentage of people in mid-labor career in the comuna (i.e., between 35 and 45 years old), a variable indicating the share of people at the moment of the life cycle supposedly most favorable for starting the own business. Results for the U.S. indicate that this variable is positively related to local business start-up rates (Reynolds et al., 1995). We also include a Herfindhal index of economic diversification, based on workers occupations according to the one-digit International Standard Classification of Occupations (ISCO). It ranges from zero (total specialization) to 100 (perfect diversification) and it is intended at capturing productivity and innovation-enhancing diversification (also called Jacobs) externalities (see Glaeser et al., 1992). Finally, the proportion of women in the population, since women have been regarded as economically and institutionally limited to self-employment (Hundley, 2001).

#### 5. RESULTS

#### Exploratory Analysis

Figure 5 depicts the three main relationships derived from the theoretical model. Panel (a) shows a weak but positive association between changes in the market potential

<sup>&</sup>lt;sup>14</sup>http://si3.bcentral.cl/Siete/secure/cuadros/home.aspx Last accessed: March 4, 2014.

<sup>&</sup>lt;sup>15</sup>http://www.vialidad.cl/productosyservicios/paginas/distancias.aspx Last accessed: December 19, 2013.



FIGURE 5: Changes in Market Potential, Wages, and Entrepreneurship Rates in Chilean *Comunas* (Logs).

and changes in self-employment rates in Chilean comunas ( $\rho = 0.119$ ). This simple correlation is small, although not inconsistent, when compared to the predictions of previous NEG models of entrepreneurship (Sato et al., 2012; Modrego et al., 2014) and also when compared to empirical results from several countries pointing at a robust positive correlation between agglomeration and entrepreneurial activity (Audretsch and Fritsch, 1994; Sato et al., 2012).

Panel (b) in turn, indicates a virtually nonexistent simple correlation between wage variations and changes in self-employment rates ( $\rho = -0.018$ ), which goes against the mechanism underlying most occupational choice models and vast empirical evidence from different countries (Parker, 1996; Belso Martínez, 2005). On the other hand, Panel (c) shows a strong positive association between changes in market potential and changes in wages ( $\rho = 0.736$ ). This result is consistent with hypothesis 1 above (that regional wages are increasing in market potential) and with empirical evidence in countries as diverse as the U.S. (Hanson, 2005), Italy (Mion, 2004), and China (Hering and Poncet, 2010).

Taken as a whole, these three pieces of information suggest that the simple correlations between regional self-employment rates, market potential and wages may be obscured by their mutual interdependencies. Market potential actually seems to exert part of its influence over regional self-employment rates indirectly through wages, what possibly explains the nil (instead of strongly negative) relationship between changes in wages and entrepreneurship rates in Figure 5(b). Further, if the (negative) opportunity cost-effect driven by higher wages offsets the (positive) profit-effect driven by spatial demand linkages, as the model above emphasizes, one would expect patterns similar to those in Figure 5.

© 2016 Wiley Periodicals, Inc.

These underlying trade-offs can be better assessed through estimation of regression parameters reflecting the *ceteris-paribus* partial effect of each covariate on the dependent variable. Along with parameter restrictions (26a) and (26b), this is a demanding statistical test of the proposed theory. We now present estimates of our econometric specification.

#### Dynamic Panel Estimates

As noted above, the observations are irregularly spaced (with two- or three-year intervals, depending on the period). This varying interval length introduces several problems in the context of dynamic panel models (see Millimet and McDonough, 2013). Millimet and McDonough (2013) evaluate a range of estimators for panels with irregular spacing, including the A-B estimator, and conclude that they are all inconsistent in presence of endogenous regressors (see table 3 in Millimet and McDonough, 2013). Wages and the market potential function are jointly determined in the theoretical model proposed here, so their exogeneity in the estimated model would be unwarranted. For these reasons, we estimate the dynamic panel equation (25) for two regularly spaced period: 1990–2000 (two years between rounds) and 2000–2009 (three-year intervals). Nevertheless, we also provide estimates for the whole 1990–2011 period with irregular intervals, while acknowledging the inconsistency of the A-B estimator in this case, to assess the implications for our results of not considering the unequal spacing between time spans.

Following the standard practice in applied studies using dynamic panel estimation, we use second-order lags in levels and differences to instrument the endogenous variables. In particular, we use lagged and twice-lagged *differences* of the market potential index and wages, and the second-order lags of the dependent variable in *levels* (which means, given the time spacing of the panel, entrepreneurship rates of the previous four or six years depending on the estimation period). One notes that the market potential measure varies within each group by considerably less than the measure's within-group averages vary across the panel; and so one might suspect some persistence to the between-group endogeneity of the lagged levels of the variable as instruments. However, the underlying conceptual argument for persistence of the market potential variable is unclear in the case of Chile: Hanson's (2005) market potential function is a combination of rapidly changing variables associated with rapid but geographically irregular economic development. Rapid household income growth has been geographically widespread but unevenly distributed (Modrego and Berdegué, 2015), and so has been the case of rapidly growing wages (Beyer and Le Foulon, 2002; Paredes, 2013) and heterogeneously growing housing prices (López and Aroca, 2012). Moreover, if persistence were a potential drawback to the use of lagged levels of the variable, lagged differences, while resolving the persistence problem, would nevertheless be small with little variation and explanatory power. Instead, we find that the best instruments appear to be lagged *differences*, not levels, suggesting that the within-group persistence of the market potential variable is not a serious concern with respect to the use of the Arellano-Band instrumenting procedure.

The instrumentation strategy was supported by the results of the well-known Hansen test of overidentifying restrictions and also by the "Difference-in-Hansen" specification test of exogeneity of subsets of instruments (see Arellano and Bond, 1991). The identifying assumption of absence of serial correlation of errors in the model in levels (equation (24)) was tested with the use of the Arellano and Bond (1991) second-order serial correlation test of errors in the differenced specification. These tests, along with consistence with parameter restrictions given by the theory, were the criteria to select the use of lagged levels of the dependent variable and lagged *differences* of the endogenous regressors as the best available instrumentation strategy.

Table 1 summarizes the regression results of the GMM estimation of the model (25). In columns (1) and (2), we present the results for the period 1990–2000, including observations for 143 comunas in the estimation sample. In columns (3) and (4), we present the results for the period 2000–2009, where the number of survey rounds falls from six to four, although the number of comunas in the estimation sample increases to 300, because of the inclusion of a large number of smaller municipalities not sampled during the 1990s. In columns (5) and (6), the results are for the whole 1990–2011 period, what yields 324 comunas in the estimation sample. Columns (1), (3), and (5) report the estimates for the base model, that is, without including additional regressors other than those implied by the relationship in equation (23). In columns (2), (4), and (6), we include the additional controls. Both the Hansen J and the Arellano-Bond serial correlation tests support model specification in our four main variants of model (25) in columns (1)-(4). The Moran I test fails to reject the null of absence of spatial correlation of errors in four out of six specifications. Comparing the results for the cases that reject and that do not reject, one cannot see qualitative differences in the point estimates or their standard errors as to suspect of biases or important efficiency problems.<sup>16</sup>

For the first estimation period, the base model (column 1) yields an estimated elasticity of self-employment rates with respect to market potential of 0.52, which is highly significant even with Windmeijer-corrected standard errors. This value is around twice as large as the nonlinear, two-step GMM estimates in Modrego et al. (2014) for the elasticity of the number of firms per capita with respect to the market potential measure (between 0.23 and 0.25). The estimated elasticity of the self-employment rate with respect to domestic wages (-0.78 in column 1 of Table 1) is in line with theoretical expectations and significant at conventional levels. Furthermore, restriction (26b)—that casts the elasticity of substitution as the negative of the ratio between the marginal effects of wages and market potential—yields a value of around 1.5, very close to the *a priori* value (1.6) used in the MP calculation in order to avoid the nonlinearity of the parameters in estimation. This comparison gives further support to our estimates, since the *a priori* value of  $\sigma$  was obtained for a model of the number of firms per capita, a metric of business activity closely related, but certainly different, to the dependent variable in this present analysis. Moreover, these previous estimates were based on a different dataset and a different estimation period (2005–2010).

Finally, the autoregressive term is positive and statistically significant. The estimated coefficient lays between the downward-biased fixed-effect estimation and the upward-biased OLS in levels, both unreported, which serves as an additional specification test (see Bond, 2002).<sup>17</sup> The positive and significant coefficient for the lag dependent variable would confirm the importance of dynamic adjustments in local entrepreneurship rates. In particular, this result supports for the Chilean case the existence of a path-dependent local entrepreneurship dynamics, a pattern also documented for business start-up rates in Sweden (Andersson and Koster, 2011). Along with the fairly significant effect of our two main explanatory variables (wages and MP) the regressions results in column (1) point at spatiotemporal persistence of local entrepreneurship in the country (Fritsch and Mueller, 2007; Andersson and Koster, 2011).<sup>18</sup>

The inclusion of additional regional controls (column 2) brings no major qualitative changes in the results compared to those in column (1). Estimated MP and wage

 $<sup>^{16}</sup>$ We are grateful to an anonymous reviewer for suggesting the application of this test.

<sup>&</sup>lt;sup>17</sup>Available upon request

<sup>&</sup>lt;sup>18</sup>The reader should be careful of not overinterpreting this parameter since there are could be many reasons simultaneously operating for this inertia, reasons we are not trying to explain here. We thank an anonymous reviewer for remarking this point.

	TABLE 1: 1	<b>Estimation</b> Resu	ilts for Model (25	).		
Parameters (Standard Errors) <sup>1</sup>	(1)	(2)	(3)	(4)	(2)	(9)
Lag log entrep. rate	$0.250^{***}$	$0.159^{**}$	0.130	0.042	$0.130^{**}$	0.066
	(0.082)	(0.079)	(0.083)	(0.077)	(0.052)	(0.057)
Log Market Potential	$0.515^{**}$	$0.567^{***}$	$0.295^{**}$	$0.408^{***}$	$0.311^{***}$	$0.329^{***}$
	(0.205)	(0.188)	(0.140)	(0.131)	(0.109)	(0.113)
Log wage	$-0.782^{**}$	$-0.763^{**}$	$-0.404^{**}$	$-0.435^{**}$	$-0.428^{***}$	$-0.425^{***}$
	(0.348)	(0.308)	(0.197)	(0.181)	(0.154)	(0.151)
Log share with university studies		-0.078		-0.094		-0.041
		(0.148)		(0.060)		(0.049)
Diversification index		$-2.616^{**}$		0.057		0.068
		(1.047)		(0.328)		(0.129)
Log share of women		0.350		-0.021		0.190
		(0.387)		(0.286)		(0.238)
Log share in mid-labor career		-0.049		-0.136		-0.089
		(0.129)		(0.087)		(0.067)
Implicit Parameter						
Sigma	$1.517^{***}$	$1.346^{***}$	$1.370^{***}$	$1.065^{***}$	$1.375^{***}$	$1.291^{***}$
	(0.402)	(0.461)	(0.391)	(0.356)	(0.266)	(0.277)
Statistics						
Ν	361	361	576	574	1,440	1,435
Global significance <i>P</i> -value	0.003	0.030	0.040	0.015	0.000	0.000
<i>P</i> -value serial correlation test (1st or.)	0.000	0.000	0.000	0.000	0.000	0.000
<i>P</i> -value serial correlation test $(2nd \text{ or.})^2$	0.930	0.417	n.a.	n.a.	0.247	0.196
$\mathbf{Instruments set}^3$	(1)	(2)	(1)	(2)	(1)	(2)
<i>P</i> -value Hansen/Sargan J	0.343	0.400	0.351	0.511	0.311	0.325
Moran's I of errors (P-value)	0.315	0.298	0.000	0.000	0.301	0.281
Estimation period	1990-2000	1990 - 2000	2000 - 2009	2000 - 2009	1990-2011	1990 - 2011
Notes: <sup>1</sup> All specifications include year dumr (including employers and self-employed), lagged lag entrepreneurship rates (including employer lagged difference of additional controls. Depende	mies. Windmeijer-corr l difference of the marl 's and self-employed), ent variable: log rates	scted standard error tet potential functio lagged difference of of employers + self-	s. <sup>2</sup> n.a. = not availab n and lagged and twi the market potentia employed in Chilean	le. <sup>3</sup> Instrument set (1 ce-lagged differences 1 function, lagged an comunas.	): Second lag entrepr of wages. Instrumen d twice-lagged differ	eneurship rates t set (2): Second ences of wages,

© 2016 Wiley Periodicals, Inc.

elasticities are very similar and still significant, but the coefficient for the lagged entrepreneurship rates decreases substantially (0.16). The implicit elasticity of substitution ( $\sigma$ ) remains above one (1.3), highly significant and with a 95 percent confidence interval containing the *a priori* value used in constructing the MP variable. Notably, additional controls provide no explanatory power, with the exception of the economic diversity index, showing a negative coefficient that suggests specialization (*Marshallian*) instead of diversification externalities. Still, the Wald test fails to reject the joint insignificance of these four additional variables (P = 0.17). Moreover, the sign of the education parameter, as well as those for the share of women and workers in mid-career are all contrary to the initial expectations, although highly not significant individually.

Results for the 2000–2009 period in columns (3) and (4) preserve the main conclusions obtained for the 1990s. First, the estimates confirms the significant positive and negative effect of the market potential variable and wages respectively, either with (column 3) or without (column 4) the inclusion of additional controls. The elasticity of entrepreneurship rates with respect to the market potential variable is now much closer to estimates by Modrego et al. (2014), which were obtained for a closer period (2005–2010). Comparing both specifications across periods (i.e., columns 1 with 3 and 2 with 4), the implicit elasticities of substitution are now slightly lower, around 1.3 and 1.1 respectively, but still highly significant and with a 95 percent confidence interval that contains the *a priori* value in both cases. Again, additional controls are not statistically significant, not even the diversification index. The most notable change is, nevertheless, the considerable reduction in the estimated coefficient for the lagged dependent variable, which is now not significant. This is to some extent an expected result, as persistence should decrease as the time interval between observations grows. Nevertheless, the sharp reduction and the loss of significance of the autoregressive coefficient in columns (3) and (4) is notable, because the length of the intervals for the 2000–2009 period is only one year longer than that of 1990–2000. Finally, although the Hansen J test supports the validity of the instruments, due to the small number of periods we were unable to perform the A-B test of second-order serial correlation, so we cannot fully dismiss the possibility of specification errors.

Finally, columns (5) and (6) report our (theoretically biased) estimates for the whole period. The main conclusion is that, given our main hypotheses, not considering the variable interval length for the data brought no serious consequence given the main hypotheses in this particular application. As expected, the autoregressive parameters lay in between those obtained for the two- and three-years intervals, which in this case implies inconclusive results in regard of entrepreneurship persistence. More important, for our main two variables, the expected sign and significance is preserved. Parameter restriction (26b) still holds and remains very similar (1.3-1.4) to the estimates for the other periods and to the *a priori* value. Finally, comparing the results in columns (1), (3), and (5) with those in columns (2), (4), and (6), one can only come to the conclusion that it is hard to anticipate the effect of omitting the varying-time interval in the case of the nonautoregressive covariates. As shown in Millimet and McDonough (2013), such effect will in general depend on their own dynamic behavior.

#### Some Robustness Checks<sup>19</sup>

The complexity of entrepreneurship as an economic phenomenon always makes the selection of an empirical metric a difficult task (see for instance Glaeser and Kerr, 2009;

 $<sup>^{19} {\</sup>rm This}$  subsection was largely motivated by the comments of one  $J\!RS$  reviewer to whom we are very grateful.

<sup>© 2016</sup> Wiley Periodicals, Inc.

Parker, 2009). In this particular case, it could be argued that the share of business owners in a municipality could be a better empirical counterpart for the entrepreneurship rates in the theoretical model. Therefore, we re-estimate model (25) by restricting the dependent variable to a measure of the share of nonfarm employers only.

Table 2 summarizes the results obtained using this employer-only alternative dependent variable. Comparing the results in Tables 1 and 2, we conclude that the main relationships in this study are still robust to the redefinition of entrepreneurship. In particular, coefficients associated to the market potential variable remain positive and significant in the six model specifications, although the rate of employers proves to be more sensitive to changes in market potential than the rate as measure by employers plus self-employed. Similarly, the coefficient on wages remains larger than the elasticity with respect to the MP function in the six model specifications. Therefore, the implicit elasticity of substitution (26b) remains greater than one, and actually within the same range as those reported in Table 1. The most notable change in this set of estimates is the small and insignificant parameter for the autoregressive term, even for the 1990–2000 period. This last result indicates that there is little or no evidence of persistence in local entrepreneurship in Chile when measured by the rate of employers in the labor force.

Taking together, the results in Tables 1 and 2 are theoretically consistent and statistically robust with respect to model specification and estimation period. The results confirm the positive partial effect of the MP and the negative effect of domestic wages on municipal self-employment rates in the country. More importantly, in all the estimations the implicit elasticity of substitution is consistent with theoretical expectations (greater than one), and not substantially different, statistically speaking, from the previously estimated value for Chile obtained from a nonlinear model (Modrego et al., 2014).

Second, we performed further robustness checks by varying the *a priori* elasticity of substitution in the computation of the market potential variable. For values of  $\sigma = [2, 3, 4]$ , the six specifications in Tables 1 and 2 yielded partial elasticities of regional entrepreneurship rates with respect to market potential and wages according to model's predictions and individually significant at least at the 10 percent. Furthermore, the parameter restriction (Equation 26b) is verified and yielded intervals that at standard confidence levels contain the *a priori* value.<sup>20</sup>

Third, modifiable aerial unit considerations are presumably important, as *comunas* could not be the most accurate representation of Chilean local economies. We re-estimated the model at the level of 44 provinces, which means with less than 200 and 100 observations for each subperiod respectively.<sup>21</sup> Although in this case finite sample biases of the A-B estimator impedes being confident about the results (nor one can expect too many significant parameters with the Windmeijer correction), point estimates proved to be qualitatively similar to municipal estimates for the model with the base specification (columns 1, 3, and 5 of Table 1). In the case of the model of entrepreneurship measured as the share of employers only, in general, it also supports the model's implication for both with and without additional controls for the periods 1990–2011 and 2000–2009 (columns 3–6 of Table 2); although the base specification for the period 1990–2011 yielded a parameter restriction (26b) lower than one.

Fourth, although officially declared as representative for "self-represented comunas" (Mideplan, 2005), CASEN survey has been criticized for providing inaccurate estimates at the municipal level, particularly for smaller comunas (see for instance Agostini and Brown, 2011). For this reason, we ran the regressions by weighting observations by the

<sup>&</sup>lt;sup>20</sup>Available upon request.

<sup>&</sup>lt;sup>21</sup>Available upon request.

<sup>© 2016</sup> Wiley Periodicals, Inc.

	TABLE 2:	Estimation Resu	ults for Model (25			
Parameters (Standard Errors) <sup>1</sup>	(1)	(2)	(3)	(4)	(2)	(9)
Lag log entrep. rate	0.005	0.024	-0.010	0.006	0.037	0.014
	(0.065)	(0.067)	(0.059)	(0.065)	(0.045)	(0.048)
Log market potential	$2.136^{***}$	$2.097^{***}$	$0.846^{**}$	$0.975^{**}$	$0.915^{***}$	$0.979^{***}$
1	(0.688)	(0.735)	(0.425)	(0.482)	(0.277)	(0.316)
Log wage	$-3.544^{***}$	$-3.275^{***}$	$-1.164^{*}$	$-1.432^{*}$	$-1.227^{***}$	$-1.044^{**}$
	(1.089)	(1.082)		(0.782)	(0.435)	(0.440)
Log share with university studies		0.015		-0.018		-0.033
		(0.439)		(0.312)		(0.179)
Diversification index		-2.124		-1.374		-0.429
		(3.443)		(2.366)		(0.488)
Log share of women		-0.955		0.043		0.350
		(0.934)		(1.404)		(0.675)
Log share in mid-labor career		-0.126		0.277		0.077
		(0.379)		(0.308)		(0.221)
Implicit Parameter						
Sigma	$1.659^{***}$	$1.561^{***}$	$1.376^{***}$	$1.469^{**}$	$1.341^{***}$	$1.066^{***}$
	(0.248)	(0.321)	(0.435)	(0.624)	(0.293)	(0.325)
Statistics						
Ν	327	327	479	479	1,179	1,179
Global significance <i>P</i> -value	0.001	0.007	0.000	0.000	0.000	0.000
<i>P</i> -value serial correlation test (1st or.)	0.001	0.002	0.000	0.000	0.000	0.000
<i>P</i> -value serial correlation test $(2nd \text{ or.})^2$	0.256	0.217	n.a.	n.a.	0.533	0.523
$Instruments set^3$	(3)	(4)	(3)	(4)	(3)	(4)
<i>P</i> -value Hansen/Sargan J	0.218	0.682	0.417	0.327	0.152	0.280
Moran's I of errors ( <i>P</i> -value)	0.629	0.632	0.000	0.000	0.242	0.241
Estimation period	1990-2000	1990-2000	2000-2009	2000 - 2009	1990-2011	1990 - 2011
<i>Notes:</i> <sup>1</sup> All specifications include year dumn (including employers only), lagged and twice-lag Second lag entrepreneurship rates (including em waors laccod difference of additional controls F	mies. Windmeijer-corn gged differences of th aployers only), lagged Denendent variable. Ic	ected standard error e market potential f and twice-lagged dif or rates of employers	s. <sup>2</sup> n.a. = not availab unction and lagged a ferences of the marke in Chilean commas	le. <sup>3</sup> Instrument set (5 nd twice-lagged diffe it potential function,	): Second lag entrepo rences of wages. Ins lagged and twice-lag	eneurship rates trument set (4): ged difference of

inverse of the number of households surveyed in each comuna in each CASEN round.<sup>22</sup> Results for the two subperiod remains the same. For the whole period, estimates equivalent to columns 5 and 6 of Table 1 and column 6 of Table 2 delivered implicit elasticities of substitution lower than 1 (although close).

Fifth, to analyze the potential role of the particular geography of extreme regions of the country, we ran the regressions excluding comunas in the regions of Arica and Parinacota, Antofagasta and Atacama (extreme north) and Aysén and Magallanes (extreme South). As correctly observed by one of the reviewers, these are low-density areas with comunas that are extremely separated from each other, and where the MP could well be given by the comuna itself. Results did not change in any significant way compared to the full-sample models.<sup>23</sup>

Finally, since our empirical proxy of municipal MP's does not consider own incomes for all but 51 comunas sampled during the whole 1990–2011 period, we reestimated the model with only these 51 comunas. Due to the small sample size, we conducted this exercise only for the 1990–2011 period. Results remained qualitatively unaltered.<sup>24</sup>

Taken as a whole, the empirical evidence provides a consistent support to the locationdriven trade-off between profit effects and opportunity-costs effects conditioning the regional supply of entrepreneurs in Chile. Consequently, the spatial occupational choice model proposed here provides a convincing explanation to the weak spatial association between market potential and entrepreneurship rates in Chile, and perhaps for other developing countries showing a similar pattern. One may argue that given the large value used for the distance decay parameter in the computation of the market potential variable, our results could be observationally equivalent to the effects of domestic market size and/or urbanization externalities in domestic markets. We believe that this is not the case here, as the market potential was built taking into account the distance to 51 major municipalities, and therefore, for most observations in the sample, home-market effects are not included.

### 6. CONCLUSION

The development of spatial economic models and methods has increased our understanding of the role of spatial demand linkages and localization externalities in shaping the geographic distribution of entrepreneurial activity. To date, however, most studies have fallen short in crafting microeconomic foundations for the role of location in the individual's decision to become an entrepreneur. This paper aims at contributing to such a line of research.

The proposed model is one of the few NEG-based framework that makes explicit the trade-offs driven by market potential in the occupational choice of entrepreneurship versus wage employment. Higher MP has a positive or pull effect on entrepreneurship through providing market opportunities for business profits, but at the same time, it has a negative influence through its positive effect on wages, the opportunity cost of being an entrepreneur. In the long run, differences in entrepreneurship rates across regions, are given by differences in average fixed costs across firms, which in turn depend on differences in the distribution of entrepreneurial ability. The effects of economic geography on entrepreneurship are thus ultimately exerted by shaping migration patterns across

<sup>&</sup>lt;sup>22</sup>Available upon request.

<sup>&</sup>lt;sup>23</sup>Available upon request.

<sup>&</sup>lt;sup>24</sup>Available upon request.

<sup>© 2016</sup> Wiley Periodicals, Inc.

regions. Overall, sorting matters (Combes et al., 2008) and models of agglomeration with heterogeneous agents (see for instance Behrens and Robert-Nicoud, 2015) are a promising avenue to build structured hypotheses regarding the role of regional market potential on the location of entrepreneurial individuals.

The model presented in this paper makes a twofold contribution. For the spatial economics literature, it introduces explicitly entrepreneurs' heterogeneity as another factor that, along with agglomeration externalities, can determine regional differences in entrepreneurship rates, average firm size, average firm costs, and labor productivity. The implication is that *everything else equal*, locations with better market access allow less cost-efficient (able) firms (entrepreneurs) to thrive. For the occupational choice literature, the model addresses the effect of location on the two key economic variables that determine the selection into entrepreneurship: entrepreneurial profits and wages.

The model proposed here inherits many of the rigidities of the standard spatial Dixit-Stiglitz framework, especially its structure of monopolistic competition with constant mark-up over marginal labor costs. It is this structure what leads to the result that, in equilibrium, the profit and opportunity cost-effect exactly offset each other. But at the same time, it is flexible enough to allow different equilibrium results. For instance, one can relax some simplifying assumptions, such as managerial ability affecting fixed costs only, or one can introduce additional elements, such as nonpecuniary motivations to choose entrepreneurship or spatial differences in start-up costs. Such extensions could well account for different geographic patterns perhaps observed in other countries in spatially-explicit occupational choice settings.

In the case of Chile, our simplifying assumptions allow us to propose an econometric specification suitable for testing the countervailing effects of market potential on entrepreneurship with the limited data at hand, which is our main motivation given the weak spatial association between these variables in Chile. The results strongly support the formal mechanics of the model, at least as a long-run equilibrium relationship towards which Chilean local economies move. Further evidence seems necessary to establish a broader empirical validation of the model in other national contexts, and in particular, to test for potential differences between developed and developing countries, where drivers of local entrepreneurship likely differ.

Finally, this paper has left open some interesting questions that deserve more theoretical work as well as empirical evidence. For instance, the model's equilibrium results would likely differ in conditions where long-run interregional migration flows might be restricted. It would be interesting to explore the long-run equilibrium with more explicit treatment of migration costs, allowing a correlation between such cost and the individual's entrepreneurial ability. This is a relevant question, because migrants are potentially different in their managerial skills and are perhaps differentially able to take advantage of place-specific assets and resources. Also, it would be important to extend the model in order to incorporate the role of amenities in explaining the spatial variation of entrepreneurship, as it has been done in some new NEG models (Südekum, 2008) and urban economic models of entrepreneurship (Glaeser, Kerr, and Ponzetto, 2010; Glaeser, Rosenthal, and Strange, 2010).<sup>25</sup> Finally, it is possible to incorporate and test for different mechanisms behind the spatial variation of entrepreneurship induced both by opportunity and by necessity (e.g., Stephens et al., 2013).

 $<sup>^{25}</sup>$ We thank an anonymous reviewer for suggesting this extension.

#### APPENDIX 1.

#### PROOF OF HYPOTHESES 1 AND 2

Totally differentiating (7) and rearranging terms yields

(A.1) 
$$W\frac{dz_r}{dz_r} + \sigma \frac{dw_r}{w_r} = \frac{dMP_r}{MP_r},$$

with  $W = -\gamma \frac{(\phi M P_r w_r^{-\sigma} - 1)}{\phi M P_r w_r^{-\sigma}} < 0$ , since the marginal entrepreneur makes nonnegative profits. And differentiating (11) and rearranging terms yields

(A.2) 
$$\sigma \frac{dw_r}{w_r} + Z \frac{dz_r}{z_r} = \frac{dMP_r}{MP_r},$$

with  $Z = \frac{w_r^r z_r}{\omega M P_r} [\frac{F(z_r)}{(1-F(z_r))^2} - \overline{k}'(z_r)] > 0$ , since  $F'(z_r) > 0$  and  $\overline{k}'(z_r) < 0$ . Stacking (A.1) and (A.2) in a displacement matrix:

$$\begin{bmatrix} W \sigma \\ Z \sigma \end{bmatrix} \begin{bmatrix} d \ln z_r \\ d \ln w_r \end{bmatrix} = \begin{bmatrix} 1 \\ 1 \end{bmatrix} d \ln MP_r.$$

Rearranging gives:

$$\begin{bmatrix} \frac{d \ln z_r}{d \ln MP_r} \\ \frac{d \ln w_r}{d \ln MP_r} \end{bmatrix} = \begin{bmatrix} W \sigma \\ Z \sigma \end{bmatrix}^{-1} \begin{bmatrix} 1 \\ 1 \end{bmatrix} = \Omega^{-1} \begin{bmatrix} 1 \\ 1 \end{bmatrix},$$

$$\begin{split} \Omega^{-1} &= \frac{1}{|\Omega|} \begin{bmatrix} \sigma & -\sigma \\ -Z & W \end{bmatrix}, \text{ which implies} \frac{d \ln z_r}{d \ln M P_r} = 0, \text{ which proves hypothesis 2.} \\ &\text{ In addition, } |\Omega| = \sigma(W - Z), \text{ so} \end{split}$$

(A.3) 
$$\frac{d\ln w_r}{d\ln MP_r} = \frac{1}{\sigma}$$

which proves hypothesis 1.

#### APPENDIX 2.

#### WELFARE ANALYSIS OF THE SHORT-RUN EQUILIBRIUM

Differentiating the profit equation (5), using (A.3) and substituting terms:

(B.1) 
$$\frac{d\pi_{ir}}{dMP_r} = \frac{\phi MP_r w_r^{1-\sigma}}{MP_r} + \frac{(1-\sigma)\phi MP_r w_r^{1-\sigma} - w_r \tau \theta_i^{-\gamma}}{w_r} \frac{dw_r}{dMP_r} = \frac{\pi_{ir}}{MP_r} \frac{1}{\sigma}$$

Equations (A.3) and (B.1) mean that all participants gain proportionally the same from an increase in the regional MP:  $\frac{d \ln(\pi)_{lr}}{d \ln(MP)_r} = \frac{d \ln(w)_r}{d \ln(MP)_r} = \frac{1}{\sigma}$ . To analyze the allocation of surplus in aggregate terms, let  $\Pi_r$  be the total surplus

To analyze the allocation of surplus in aggregate terms, let  $\Pi_r$  be the total surplus in the region which comprises the sum of total entrepreneurial  $(\Pi_r^e)$  and total wage  $(\Pi_r^w)$  rents:

$$\Pi_r = L_r(\int_0^{z_r} w_r f(\theta) d\theta + \int_{z_r}^{+\infty} \pi_{ir} f(\theta) d\theta) = \Pi_r^w + \Pi_r^e.$$

Using (A.3) and (B.1), the effect of a shock in the regional market potential on the total surplus is  $\frac{d\Pi_r}{dMP_r} = \frac{w_r}{\sigma MP_r} l_r + \frac{\overline{\pi}_r}{\sigma MP_r} e_r$ , where  $\overline{\pi}_r$  indicates average business profits.

© 2016 Wiley Periodicals, Inc.

The relative additional rents triggered by the increase in the market potential are

$$rac{\left(rac{d\Pi_r^e}{dMP_r}
ight)}{\left(rac{d\Pi_r^w}{dMP_r}
ight)} = rac{\overline{\pi}_r}{w_r}rac{e_r}{l_r} = rac{\overline{\pi}_r}{\pi(z_r)}rac{e_r}{l_r},$$

that is, the relative gains of both groups depend only on their relative sizes.

#### REFERENCES

- Acs, Zoltan J. and Catherine Armington. 2004. "Employment Growth and Entrepreneurial Activity in Cities," *Regional Studies*, 38(8), 911–927.
- Agostini, Claudio and Philip Brown. 2011. "Cash Transfers and Poverty Reduction in Chile," Journal of Regional Science, 51(3), 604–625.
- Andersson, Martin and Sierdjan Koster. 2011. "Sources of Persistence in Regional Start-Up Rates—Evidence from Sweden," Journal of Economic Geography, 11, 179–201.
- Arellano, Manuel and Stephen R. Bond. 1991. "Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations," *Review of Economic Studies*, 58, 277–297.
- Audretsch, David B. and Michael Fritsch. 1994. "The Geography of Firm Births in Germany," Regional Studies, 28(4), 359–365.
- Audretsch, David B. and Max Keilbach. 2004. "Entrepreneurship Capital and Economic Performance," Regional Studies, 38(8), 949–959.
- Behrens, Kristian and Frédéric Robert-Nicoud. 2015. "Agglomeration Theory with Heterogeneous Agents" in Gilles Duranton, Vernon Henderson, and William C. Strange (eds.), *Handbook of Regional and Urban Economics*, Vol. 5A. North Holland: Elsevier.
- Belso Martínez, José A. 2005. "Equilibrium Entrepreneurship Rate, Economic Development and Growth. Evidence from Spanish Regions," *Entrepreneurship & Regional Development*, 17(2), 145–161.
- Beyer, Harald and Carmen Le Foulon. 2002. "Un Recorrido por las Desigualdades Salariales en Chile," Estudios Públicos, 85, 139–175.
- Blanchflower, David G. and Andrew J. Oswald. 1998. "What Makes an Entrepreneur?" Journal of Labor Economics, 16, 26–60.
- Boettke, Peter J. and Cristopher J. Coyne. 2003. "Entrepreneurship and Development: Cause or Consequence?" Advances in Austrian Economics, 6, 67–87.
- Bond, Stephen R. 2002. "Dynamic Panel data Models: A Guide to Micro Data Methods and Practice," Portuguese Economic Journal, 1, 141–162.
- Bosma, Niels and Jonathan Levie. 2010. Global Entrepreneurship Monitor, 2009 Global Report. Wellesley: Babson College.
- Carree, Martin A. and Roy Thurik. 2003. "The Impact of Entrepreneurship on Economic Growth," in Zoltan Acs and David B. Audretsch (eds.), *Handbook of Entrepreneurship Research*. Boston: Kluwer Academic Publishers.
- Combes, Pierre-Philippe, Gilles Duranton, and Laurent Gobillon. 2008. "Spatial Wage Disparities: Sorting Matters," Journal of Urban Economics, 63(2), 723–742.
- Davidson, Per and Johan Wilklund. 1997. "Values, Beliefs and Regional Variations in New Firm Formation Rates," Journal of Economic Psychology, 18, 179–199.
- De Blasio, Guido and Giorgio Nuzzo. 2010. "Historical Traditions of Civicness and Local Economic Development," Journal of Regional Science, 50(4), 833–857.
- Evans, David S. and Bojan Jovanovic. 1989. "An Estimated Model of Entrepreneurial Choice under Liquidity Constraints," The Journal of Political Economy, 97(4), 808–827.
- Fingleton, Bernard. 2006. "The New Economic Geography Versus Urban Economics: An Evaluation Using Local Wage Rates in Great Britain," Oxford Economic Papers, 58, 501–530.
- Fingleton, Bernard and Manfred M. Fischer. 2010. "Neoclassical Theory versus New Economic Geography: Competing Explanations of Cross-Regional Variation in Economic Development," Annals of Regional Science, 44, 467–491.
- Fritsch, Michael and Pamela Mueller. 2007. "The Persistence of Regional New Business Formation Activity over Time—Assessing the Potential of Policy Promotion Programs," *Journal of Evolutionary Economics*, 17, 299–315.
- Forslid, Rikard and Gianmarco I.P. Ottaviano. 2003. "An Analytically Solvable Core-Periphery Model," Journal of Economic Geography, 3, 229–240.
- Fujita, Masahisa, Paul Krugman, and Anthony Venables. 1999. The Spatial Economy: Cities, Regions, and International Trade. Cambridge: MIT Press.

- Glaeser, Edward and William Kerr. 2009. "Local Industrial Conditions and Entrepreneurship: How Much of the Spatial Distribution Can We Explain?" Journal of Economics and Management Strategy, 18(3), 623–663.
- Glaeser, Edward, Hedi Kallal, José Scheinkman, and Andrei Shleifer. 1992. "Growth in Cities," Journal of Political Economy, 100, 1126–1152.
- Glaeser, Edward, William Kerr, and Giacomo Ponzetto. 2010. "Clusters of Entrepreneurship," Journal of Urban Economics, 67, 150–168.
- Glaeser, Edward, Stuart S. Rosenthal, and William C. Strange. 2010. "Urban Economics and Entrepreneurship," Journal of Urban Economics, 67, 1–14.
- Goetz, Stephan J. and Anil Rupasingha. 2009. "Determinants of Growth in Non-Farm Proprietor Densities in the US, 1990-2000," Small Business Economics, 32, 425–438.
- Hanson, Gordon. 2005. "Market Potential, Increasing Returns, and Geographic Concentration," Journal of International Economics, 67(1), 1–24.
- Harris, Chauncy D. 1954. "The Market as a Factor in the Localization of Industry in the United States." Annals of the Association of American Geographers, 44, 315–348.
- Hering, Laura and Sandra Poncet. 2010. "Market Access and Individual Wages: Evidence from China," The Review of Economics and Statistics, 92(1), 145–159.
- Helpman, Elhanan. 1998. "The Size of Regions" in David Pines, Efraim Sadka, and Itzhak Zilcha (eds.), Topics in Public Economics: Theoretical and Applied Analysis. Cambridge: Cambridge University Press.
- Hundley, Greg. 2001. "Why Women Earn Less than Men in Self-Employment," *Journal of Labor Research*, 22, 817–829.
- Krugman, Paul. 1991. "Increasing Returns and Economic Geography," Journal of Political Economy, 99, 483-499.

Lazear, Edward P. 2004. "Balanced Skills and Entrepreneurship," *American Economic Review*, 94(2), 208–211. \_\_\_\_\_\_. 2005. "Entrepreneurship," *Journal of Labor Economics*, 23(4), 649–680.

- López, Esteban and Patricio Aroca. 2012. "Estimación de la Inflación Regional de los Precios de la Vivienda en Chile," El Trimestre Económico, 79(315), 601–630.
- Lucas, Robert E. 1978. "On the Size Distribution of Business Firms," Bell Journal of Economics, 9, 508-523.
- Mideplan (currently Ministerio de Desarrollo Social). 2005. "Encuestas de Caracterización Socioeconómica Nacional CASEN 2003: Marco Metodológico." Departamento de Información Social, División Social. Santiago de Chile.
- Millimet, Daniel L. and Ian, K. McDonough. 2013. "Dynamic Panel Data Models with Irregular Spacing: With Applications to Early Childhood Development," IZA Discussion Paper No. 7359. Bonn: Institute for the Study of Labor (IZA).
- Mion, Giordano. 2004. "Spatial Externalities and Empirical Analysis: The Case of Italy," Journal of Urban Economics, 56, 97–118.
- Modrego, Félix and Julio A. Berdegué. 2015. "A Large-Scale Mapping of Territorial Development Dynamics in Latin America," World Development, 73, 11–31.
- Modrego, Félix, Philip McCann, William E. Foster, and M. Rose Olfert. 2014. "Regional Market Potential and the Number and Size of Firms: Observations and Evidence from Chile," *Spatial Economic Analysis*, 9(3), 327–348.
- ———. 2015. "Regional Entrepreneurship and Innovation in Chile: A Knowledge Matching Approach," *Small Business Economics*, 44, 685–703.
- Modrego, Félix, Dusan Paredes, and Gianni Romaní. 2015. "Individual and Place-Based Drivers of Self-Employment in Chile," working paper. Antofagasta, Chile: Department of Economics, Universidad Católica del Norte.
- Murphy, Kevin, Andrei Shleifer, and Robert Vishny. 1990. "The Allocation of Talent: Implications for Growth," Quarterly Journal of Economics, 106(2), 503–530.
- Naudé, Wim, Thomas Gries, Eric Wood, and Aloe Meintjes. 2008. "Regional Determinants of Entrepreneurial Start-ups in a Developing Country," *Entrepreneurship & Regional Development*, 20(2), 111–124.
- Paredes, Dusan. 2013. "The Role of Human Capital, Market Potential and Natural Amenities in Understanding Spatial Wage Disparities in Chile," Spatial Economic Analysis, 8(2), 154–175.
- Parker, Simon C. 1996. "A Time Series Model of Self-Employment under Uncertainty," Economica, 63(251), 459–475.
- ——. 2005. "Explaining Regional Variations in Entrepreneurship as Multiple Equilibria," *Journal of Regional Science*, 45(4), 829–850.
- \_\_\_\_\_. 2009. The Economics of Entrepreneurship. Cambridge: Cambridge University Press.
- Reynolds, Paul D., Niels Bosma, and Erkko Autio. 2005. "Global Entrepreneurship Monitor: Data Collection Design and Implementation 1998–2003," Small Business Economics, 24(3), 205–231.
- Reynolds, Paul D., Brenda Miller, and Wilbur R. Maki. 1995. "Explaining Regional Variation in Business Births and Deaths: U.S. 1976–88," Small Business Economics, 7, 389–407.

Roback, Jennifer. 1982. "Wages, Rents, and the Quality of Life," Journal of Political Economy 90: 1257-1278.

Robinson, Peter B. and Edwin A. Sexton. 1994. "The Effect of Education and Experience on Self-Employment Success," *Journal of Business Venturing*, 9(2), 141–156.

- Romaní, Gianni, Miguel Atienza, and José Ernesto Amorós. 2009. "Financing Entrepreneurial Activity in Chile: Scale and Scope of the Public Support Programs," *Venture Capital: An International Journal of Entrepreneurial Finance*, 11(1), 50–70.
- Roodman, David. 2009. "How to do xtabond2: An Introduction to 'Difference' and 'System' GMM in Stata," *The Stata Journal*, 9(1), 86–136.
- Sato, Yasuhiro, Takatoshi Tabuchi, and Kazuhiro Yamamoto. 2012. "Market Size and Entrepreneurship," Journal of Economic Geography, 12, 1139–1166.
- Saxenian, Annalee. 1994. Regional Advantage: Culture and Competition in Silicon Valley and Route, Vol. 128. Cambridge, MA: Harvard University Press.
- Stephens, Heather M., Mark D. Partridge, and Alessandra Faggian. 2013. "Innovation, Entrepreneurship and Growth in Lagging Regions," *Journal of Regional Science* 53(5), 778–812.
- Südekum, Jens. 2008. "Regional Costs-of-Living with Congestion and Amenity Differences: an Economic Geography Perspective," The Annals of Regional Science, 43, 49–69.
- Van Soest, Daan, Shelby Gerking, and Frank Van Oort. 2006. "Spatial Impacts of Agglomeration Externalities," Journal of Regional Science, 46(5), 881–899.
- Windmeijer, Frank. 2005. "A Finite Sample Correction for the Variance of Linear Efficient Two-Step GMM Estimators," Journal of Econometrics, 126, 25–51.
- Wu, Jun Jie and Munisamy Gopinath. 2008. "What Causes Spatial Variations in Economic Development in the United States?" American Journal of Agricultural Economics, 90(2), 392–408.